\documentclass[12pt]{article}

\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{enumerate}

\setlength{\parskip}{0.2cm} % Space between paragraphs
\setlength\parindent{0pt} % Eliminate Paragraph Indentation

\begin{document}
\textbf{Name}: -- Your first and last name here --

\textbf{Date}: -- Today's date here as mm/dd/yyyy --

\textbf{Assignment \#}: 2

\hrulefill

\section*{Answers}

% Type your answers inside the list below. E.g., the answer to question 1 will be typed inside the first \item. [You may remove this comment from your source code file later if you want to.]
\begin{enumerate}
\item
\item
\item
\item
\item
\end{enumerate}

\end{document}
